Author’s response to reviews

Title: Association between NRGN Gene Polymorphism and Resting-State Hippocampal Functional Connectivity in Schizophrenia

Authors:

Yifan Zhang (ivanhey825@163.com)
Xiaohong Gong (gongxh@fudan.edu.cn)
Zhiyang Yin (zyyls@163.com)
Lingling Cui (13804071019@163.com)
Jian Yang (yangjian@cmu.edu.cn)
Pengshuo Wang (Kyleonaz@163.com)
Yifang Zhou (zhou060117@126.com)
Shengnan Wei (weisn1982@126.com)
Xiaowei Jiang (jxw024@163.com)
Fei Wang (fei.wang@cmu.edu.cn)
Yanqing Tang (tangyanqing@cmu.edu.cn)

Version: 3 Date: 19 Mar 2019

Author’s response to reviews:

1. We note that the current submission contains some textual overlap with other previously published works. This overlap mainly exists in the Methods section.

The overlap primarily occurs between your submission and the previously published sources:

"Alteration of cortico-limbic-striatal neural system in major depressive disorder and bipolar disorder"

"Variation in the CACNB2 gene is associated with functional connectivity of the Hippocampus in bipolar disorder"


While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. Please note that an abundance of overlapping text may preclude a given manuscript from publication. Therefore, we ask that you please re-phrase these sections to minimise overlap and prevent future delays in the editorial process for your manuscript.

We have rephrased the sections containing overlap in the revised manuscript.

2. Please note that all authors should be named by using their initials rather than full names in the Authors’ contributions section. For example, author John Smith should be represented as JS in this section. Please note that when two or more authors have the same initials they should be distinguished by designating them with numbers 1 and 2 (etc.). For example: YZ1 and YZ2. YZ1 would correspond to the author furthest up on the author list.

We have modified all authors’ names in the Authors' Contributions section in the revised manuscript.

3. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

We have erased all highlighted text in the revised manuscript.